Author’s response to reviews

Title: INTERNATIONAL palliative CARE RESEARCH PRIORITIES: A SYSTEMATIC REVIEW

Authors:

Felicity Hasson (f.hasson@ulster.ac.uk)

Emma Nicholson (emma.nicholson@ucd.ie)

Deborah Muldrew (d.muldrew@ulster.ac.uk)

Olufikayo Bamidele (Olufikayo.Bamidele@hyms.ac.uk)

Sheila Payne (s.a.payne@lancaster.ac.uk)

Sonja McIlfatrick (sj.mcilfatrick@ulster.ac.uk)

Version: 1 Date: 11 Nov 2019

Author’s response to reviews:

11th Nov 2019

BMC Palliative Care

Dear Sir/Madam,

Title: TITLE: INTERNATIONAL palliative CARE RESEARCH PRIORITIES: A SYSTEMATIC REVIEW

Please find attached a copy of the above manuscript, for consideration of publication within BMC Palliative Care, with PRISMA checklist now attached.

This work has not been published elsewhere and is not under consideration for publication elsewhere. All authors meet the criteria for authorship in the journal and all authors have contributed equally to the development of this article. All authors have seen and approved the manuscript. No conflict of interests exists for any author.
If you require any further information, please do not hesitate to contact me on email: f.hasson@ulster.ac.uk.

Yours faithfully

Dr Felicity Hasson, Senior Lecturer
Ulster University, Institute of Nursing and Health Sciences Research
Newtownabbey, Shore Road, BT37 0QB, Co. Antrim
Tel: 0044 (0) 28 90 36 6895
Email: f.hasson@ulster.ac.uk